Author's response to reviews

Title: Possible involvement of coveolin in attenuation of cardioprotective effect of ischemic preconditioning in diabetic rat heart

Authors:

  Preeti Ajmani (ajmani_preeti@rediffmail.com)
  Harlokesh Narayan Yadav (hnyadav@gmail.com)
  Manjeet Singh (amitozbaidwan@gmail.com)
  Pyare L Sharam (plsharma.isf@gmail.com)

Version: 5  Date: 8 July 2011

Author's response to reviews: see over
Dear Editor
Thank you very much for your letter of July, 7, 2011 lighting the formatting change to be made in the manuscript no 322562425173078.

I am herewith resubmitting the revised manuscript after making all the required change as detail below.

Author list: The list of author in the manuscript and in the resubmission system is now written in the same system.

Affiliation: Is now written in the full, as suggested by you.

Abstract: Is now identical in manuscript and resubmission system. This includes the headings.

References: The references are cited in numerical order. References 67, 68 and 69 are cited in the second paragraph of the discussion.

Author List: No title has been used for author.

Email-address: The e-mail addresses of all the authors are included on the title page, using the format suggested by you.

Figures: The figures are cropped as closely as possible. The title of figures has been removed from the image file. The legend and title now are the part of the manuscript after reference list. The blank pages between figure files have been removed.

Typography: The manuscript has been checked again for any typographical error.

With regard
Harlokesh N Yadav
Asst. Professor, ISF College of Pharmacy
Moga, Punjab, India.